Author's response to reviews

Title: Bloodstream infection following 217 consecutive systemic-enteric drained pancreas transplants

Authors:

Natalie R Berger (Natalie.Berger@uibk.ac.at)
Siegmund Guggenbichler (Siegmund.Guggenbichler@uibk.ac.at)
Wolfgang Steurer (W.Steurer@vodafone.de)
Christian Margreiter (Christian.Margreiter@uibk.ac.at)
Gert Mayer (Gert.Mayer@uibk.ac.at)
Reinhold Kafka (Reinhold.Kafka@uklibk.ac.at)
Walter Mark (Walter.Mark@uibk.ac.at)
Alexander R Rosenkranz (Alexander.Rosenkranz@uibk.ac.at)
Raimund Margreiter (Raimund.Margreiter@uibk.ac.at)
Hugo Bonatti (Hugo.bonatti@uklibk.ac.at)

Version: 7 Date: 21 July 2006

Author's response to reviews:

All requested changes have been made

Please remove the following text from the title page: 'ORIGINAL ARTICLE; Key words: pancreas transplantation, enteric drainage, infection, sepsis'
removed
Could you make sure that there is no unnecessary bold text in the manuscript.
changed: all changes are in different colours so they can be tracked
Conclusion - Please include a Conclusions section after Discussions or rename the Discussions section to 'Discussions and Conclusions'.
changed
After the References there should be a section entitled 'Figure legends', which contains the Figure descriptions only (not table descriptions).
changed
Next should be the Tables section with the title and description of the table included with the table. No separate table description is required.
changed